
\subsection{Barry G. Ritchie}

\noindent{\bf{Professional preparation}}
\begin{table}[ht]
\centering
\begin{tabular}{llll}
 & \underline{Institution} & \underline{Major/Conc.} & \underline{Degree, Year} \\
Undergraduate: & Appalachian State University & Physics & B.S., 1975 \\
Graduate: & University of South Carolina & Physics & M.S., 1977 \\
          & University of South Carolina & Physics & Ph.D., 1979 \\
Postdoctoral: & University of South Carolina & Physics & 1979-1983\\
              & University of Maryland       & Physics & 1983-1984\\ 
\end{tabular}
\end{table}

\noindent{\bf{Appointments}}
\begin{table}[ht]
\centering
\begin{tabular}{lll}
Professor           & Arizona State University & 1996-present\\
Senior Advisor to the Provost & Arizona State University & 2015-2016\\
Vice Provost for Academic Personnel & Arizona State University & 2012-2015\\
Interim Dean and Vice President       & Arizona State University & 2006-2007\\
Chair               & Arizona State University & 2000-2006\\
Associate Professor & Arizona State University & 1990-1996\\
Visiting Associate Professor & University of Virginia & 1990-1991\\
Assistant Professor & Arizona State University & 1984-1990\\
\end{tabular}
\end{table}

\noindent{\bf Synergistic activities}
\vskip0.15in
Technical Advisory Panel Representative, Hall B, CEBAF, 1990-92;  
Member, Board of Directors, LAMPF Users' Group, Inc., 1992-94;
LAMPF Experimental Facilities Panel member, 1990-94;
Chair, CLAS Real Photon Physics Working Group
Member, CLAS Coordinating Committee, 1995-99; 
\newline

%\vskip0.15in

%
%\noindent{\bf e) Collaborations and other affiliations}
%
%\vskip0.15in
%
%\noindent   
%
%%Reference \cite{clasMemberList} in Section D (References) lists the 
%%names of the collaborators and collaborating institutions who are 
%%associated with the CLAS Collaboration. 
%
%\input glueXMembers2016Vs.tex
%
%\input clasMembers2016Vs.tex
%
%
%\vskip0.15in
%
%\noindent{{\bf{Ph.D. supervisor:}}} F. T. Avignone, III, University of South Carolina
%
%
%\vskip0.15in
%
%
%\noindent{{\bf{Postdoctoral supervisors:}}} B. M. Preedom, University of South Carolina; P. G. Roos, University of Maryland.
%
%%
%\end{document}

\begin{bibunit}
\makeatletter
\renewcommand\@biblabel[1]{#1.}
\makeatother
\makeatletter
\renewenvironment{thebibliography}[1]
     {\list{\@biblabel{\@arabic\c@enumiv}}%
           {\settowidth\labelwidth{\@biblabel{#1}}%
            \leftmargin\labelwidth
            \advance\leftmargin\labelsep
            \@openbib@code
            \usecounter{enumiv}%
            \let\p@enumiv\@empty
            \renewcommand\theenumiv{\@arabic\c@enumiv}}%
      \sloppy
      \clubpenalty4000
      \@clubpenalty \clubpenalty
      \widowpenalty4000%
      \sfcode`\.\@m}
     {\def\@noitemerr
       {\@latex@warning{Empty `thebibliography' environment}}%
      \endlist}
\makeatother

\noindent{
\textbf{Publications over the last three years}
\phantom{~\cite{
%%Added by BGR
Fedotov:2018oan,Chetry:2018kxz,
Chandavar:2017lgs,Hattawy:2017woc,Compton:2017xkt,Ho:2017kca,Isupov:2017lnd,
%% Original list from MRD
Kunkel:2017src,Roy:2017qwv,Akbar:2017uuk,
Anisovich:2017pox,Collins:2017vev,Mattione:2017fxc,
Anisovich:2017rpe,Dugger:2017zoq,Collins:2017sgu,AlGhoul:2017nbp,
Paterson:2016vmc,Ghoul:2015ifw} as shown below.}}
\putbib[bibtexrefs2018]
\end{bibunit}



